Author's response to reviews

Title: Excessive proinflammatory cytokine and chemokine responses of human monocyte-derived macrophages to enterovirus 71 infection

Authors:

Xun Gong (gongxun8269@163.com)
Jianfang Zhou (jfz@cnic.org.cn)
Wenfei Zhu (wenfei@cnic.org.cn)
Na Liu (unali@163.com)
Jinsong Li (songpia@163.com)
Lili Li (lilili030412@163.com)
Yu Jin (jinyuldyy@163.com)
Zhaojun Duan (zhaojund@126.com)

Version: 2 Date: 18 January 2012

Author's response to reviews:

BMC Infectious Diseases
18th Jan. 2011

Dear Editor,

Re: Excessive proinflammatory cytokine and chemokine responses of human monocyte-derived macrophages to enterovirus 71 infection

Thank you very much for sending us the requirements and suggestions.

We have modified the manuscript accordingly. The details of revision were explained as following.

1) The section of “methods” between background and results in the abstract was already added in the revised manuscript.

2) As to the statement of Ethics on the project, we have amended it in the revised manuscript. Please refer to Paragraph 1 in the Methods section of the revised manuscript.

Yours sincerely,

Zhao-Jun Duan

National Institute for Viral Disease Control and Prevention, China CDC